 Vc/9-Aug-04
Week32 starting 2Aug  04

SM18 Operation Meeting (Tuesdays)

 Bench status  on Monday 02 aug04  18.15 hrs 

	TBA1
	MBAR1139
	ICS 3 Traveller after warm test
	303.6

	TBA2
	MBBL3191
	Prep 5 Pumping + Cool Down
	90.8

	TBB1
	NONE
	Bench modif to Spec SSS
	302.5

	TBB2
	MBBL2090
	PTE Min. NRJ 3KA Diode Test
	1.9

	TBC1
	MBAL3172
	PT 12 Warm Up
	301.9

	TBC2
	NONE
	Bench modif to Dipole
	302.5

	TBD1
	SSS023
	ICS 5 Traveller Last
	302.5

	TBD2
	SSS036
	Stretched Wire Warm SW11
	301.4

	TBE1
	MBBL3137
	PT 10 Min. NRJ 11850 A
	11.6

	TBE2
	_
	_
	303.1

	TBF1
	MBAR1140
	PT 6.9 Training 9
	2.35

	TBF2
	MBBL1119
	PT 2 HV Cold Test
	1.9


8/2/2004 6:11:33 PM    Temperature data are taken from TBSM18 site
Coordinator this week: Stephane ; MTM support on call :  (16-3602).

10 Magnets( 8 dipoles+ 2 SSS)   Disconnected after tests last week    ( Sat Midnight to Sat midnight)  ; + 3 Dipoles & 1 SSS that finished Training & warm up launched  but not met Sat midnight cutoff [ A1: 3177, E2: 2035 , B1:3190 , D1 : SS023 ]
· Situation From Cryogenics:  J-P Lamboy or ACR Rep.(Jerome ?)

· Modifs to C2 to take on Dipoles
· Modifs to B1 to take on SSS & later Special SSS’s
· SSS work :
· To Do Lists & SMTMS updates & simplifications
· Diagnostics & Faults chasing  
· Results & “goodness” evaluation 
Bench status  and tests for now( Monday 2 Aug/7at 18.20 hrs)  & next few days :  see Video Display

	Magnets under Test on 02-08 6:17:33 PM 

	-
	Bench
	Magnet
	Shafts
	Start
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	Last Test
Desc
	
Time
	
I
	
Temp.
	Total

	1
	TBA1
	MBAR1139 
	[image: image1.png]



	02-08 
	14 h.
	-
	-
	-
	-
	ICS 3 Traveller after warm test
	02-08
(16 : 00) 
	0 A. 
	300 K
	14 hours

	2
	TBA2
	MBBL3191 
	[image: image2.png]



	31-07 
	34 h.
	21 h.
	-
	-
	-
	Prep 5 Pumping + Cool Down
	01-08
(21 : 34) 
	0 A. 
	Cooling
	55 hours

	3
	TBB1
	NONE 
	[image: image3.png]



	02-08 
	8 h.
	-
	-
	-
	-
	Bench modif to Spec SSS
	02-08
(10 : 18) 
	0 A. 
	300 K
	8 hours

	4
	TBB2
	MBBL2090 
	[image: image4.png]



	29-07 
	63 h.
	19 h.
	26 h.
	-
	-
	PTE Min. NRJ 3KA Diode Test
	02-08
(04 : 37) 
	3000 A. 
	1.9 K
	108 hours

	5
	TBC1
	MBAL3172 
	[image: image5.png]



	26-07 
	48 h.
	71 h.
	34 h.
	22 h.
	-
	PT 12 Warm Up
	02-08
(06 : 00) 
	0 A. 
	4.4 K
	175 hours

	6
	TBC2
	NONE 
	[image: image6.png]



	02-08 
	8 h.
	-
	-
	-
	-
	Bench modif to Dipole
	02-08
(10 : 11) 
	0 A. 
	300 K
	8 hours

	7
	TBD1
	SSS023 
	[image: image7.png]



	19-07 
	98 h.
	-84 h.
	298 h.
	17 h.
	1 h.
	ICS 5 Traveller Last
	02-08
(08 : 50) 
	0 A. 
	300 K
	340 hours

	8
	TBD2
	SSS036 
	[image: image8.png]



	28-07 
	120 h.
	-
	-
	-
	-
	Stretched Wire Warm SW11
	02-08
(10 : 03) 
	0 A. 
	300 K
	120 hours

	9
	TBE1
	MBBL3137 
	[image: image9.png]



	28-07 
	58 h.
	29 h.
	43 h.
	-
	-
	PT 10 Min. NRJ 11850 A
	02-08
(13 : 40) 
	11850 A. 
	1.9 K
	130 hours

	10
	TBF1
	MBAR1140 
	[image: image10.png]



	27-07 
	69 h.
	2 h.
	74 h.
	-
	-
	PT 9 CDAP HV
	02-08
(17 : 00) 
	0 A. 
	1.9 K
	145 hours

	11
	TBF2
	MBBL1119 
	[image: image11.png]



	29-07 
	33 h.
	61 h.
	-
	-
	-
	PT 2 HV Cold Test
	02-08
(10 : 30) 
	0 A. 
	4.4 K
	94 hours

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAR1139
	[image: image12.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	2
	MBBL3191
	[image: image13.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	NONE
	[image: image14.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBBL2090
	[image: image15.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBAL3172
	11656 Amp.
(D2_U)
	12851 Amp.
([image: image16.jpg]


)
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	NONE
	[image: image17.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS023
	12287 Amp.
(QI-34)
	12616 Amp.
(QI-12)
	12686 Amp.
(QE_12)
	12845 Amp.
(QE_12)
	12855 Amp.
(QE_34)
	[image: image18.png]



	-
	-
	-
	-
	-
	-
	-

	8
	SSS036
	[image: image19.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBL3137
	11089 Amp.
(D2-L)
	12296 Amp.
(D2_U)
	[image: image20.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBAR1140
	11202 Amp.
(D2-L)
	11645 Amp.
(D2-L)
	12142 Amp.
(D2-L)
	12383 Amp.
(D1-L)
	12513 Amp.
(D1-U)
	12597 Amp.
(D2 L)
	12662 Amp.
(D1-U)
	12700 Amp.
(D2-U)
	12332 Amp.
(D2_U)
	[image: image21.jpg]



	-
	-
	-

	11
	MBBL1119
	[image: image22.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-


Results by Week for PREVIOUS week 

	

	Magnets under Test from 25-07-2004 to 31-07-2004 

	-
	Bench
	Magnet
	Shafts
	Start Date .
	Connect
Time
	Cooling 
Time
	Cold Test 
Time
	Warming 
Time
	Disconnect
Time
	End Date
	Total
	
	

	1
	TBA1
	MBAR1136 
	[image: image23.png]



	14-07-2004 
	42 h.
	34 h.
	151 h.
	17 h.
	11 h.
	25-07-2004 
	264 hours
	
	

	2
	TBA2
	MBBL3189 
	[image: image24.png]



	24-07-2004 
	46 h.
	43 h.
	65 h.
	16 h.
	9 h.
	31-07-2004 
	168 hours
	
	

	3
	TBB1
	MBBL1098 
	[image: image25.png]



	24-07-2004 
	39 h.
	-
	-
	-
	-
	26-07-2004 
	48 hours
	
	

	4
	TBB2
	MBAL3180 
	[image: image26.png]



	22-07-2004 
	28 h.
	55 h.
	32 h.
	20 h.
	3 h.
	29-07-2004 
	168 hours
	
	

	5
	TBC1
	MBAL3174 
	[image: image27.png]



	19-07-2004 
	29 h.
	60 h.
	35 h.
	12 h.
	23 h.
	26-07-2004 
	168 hours
	
	

	6
	TBC2
	SSS031 
	[image: image28.png]



	22-07-2004 
	69 h.
	24 h.
	107 h.
	10 h.
	5 h.
	31-07-2004 
	216 hours
	
	

	7
	TBD2
	SSS017 
	[image: image29.png]



	02-07-2004 
	178 h.
	147 h.
	270 h.
	15 h.
	14 h.
	28-07-2004 
	624 hours
	
	

	8
	TBE1
	MBBL2091 
	[image: image30.png]



	21-07-2004 
	16 h.
	88 h.
	21 h.
	19 h.
	25 h.
	28-07-2004 
	168 hours
	
	

	9
	TBF1
	MBBL1129 
	[image: image31.png]



	19-07-2004 
	35 h.
	66 h.
	64 h.
	23 h.
	1 h.
	27-07-2004 
	192 hours
	
	

	10
	TBF2
	MBBL2085 
	[image: image32.png]



	21-07-2004 
	8 h.
	8 h.
	142 h.
	18 h.
	1 h.
	29-07-2004 
	192 hours
	
	


+ 3 Dipoles & 1 SSS that finished Training & warm up launched  but not met Sat midnight cutoff [ A1: 3177, E2: 2035 , B1:3190 , D1 : SS023 ]

Quench Information
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAR1136
	10853 A.
(D2-L-S13)
	11213 A.
(D2-L-S12)
	11451 A.
(D2-U S13)
	11722 A.
(D2-L-S13)
	11892 A.
(D2_U_S13)
	12037 A.
(D1_U_S01)
	12193 A.
(D1_L_S13)
	12244 A.
(D2_L_S01)
	12371 A.
(D2-L-S12)
	12036 A.
(D2-L-S13)
	12363 A.
(D1-LS 13)
	[image: image33.png]



	-

	2
	MBBL3189
	11139 A.
(D2-L)
	12270 A.
(D2 U)
	[image: image34.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	MBBL1098
	[image: image35.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	MBAL3180
	8753 A.
(D1-L)
	12006 A.
(D2-U)
	[image: image36.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	5
	MBAL3174
	10739 A.
(D1_U)
	11885 A.
(D1_U)
	12300 A.
(d2_L)
	12850 A.
[image: image37.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	6
	SSS031
	12679 A.
(D1_L)
	12823 A.
(Q-E-12)
	12851 A.
[image: image38.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	7
	SSS017
	11325 A.
(Q_E 12)
	11607 A.
(QI_12)
	12095 A.
(QI-12)
	12541 A.
(QI-34)
	11431 A.
(No Data)
	12850 A.
[image: image39.jpg]



	-
	-
	-
	-
	-
	-
	-

	8
	MBBL2091
	11983 A.
(D2_U)
	12859 A.
[image: image40.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	9
	MBBL1129
	12217 A.
(D2_L-S01)
	12589 A.
(D2_U_s13)
	[image: image41.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	10
	MBBL2085
	11527 A.
(D1-U-S01)
	12533 A.
(D1-U-S01)
	[image: image42.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-


 

General Problems last week :  
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday _27_A

Wednesday_28_A
	TBC2
	Power supply on Auxiliary circuit cannot be started.


	MTM
	
	Power supply tested by AB PO in short circuit. OK

L measured on circuit CD and EF, found inductances of 5 and 10 mH instead of 18 mH and 62 mH.

	Tuesday _27_A
	TBD1
	HV Test : Report incomplete and test sequence on SS23 did not match with that used for previous SSS (SSS17).


	MTM/AB/CO
	HV
	Federico and TC informed.

Richard: Sequence of the dipole was launched by mistake. Then aborted manually and switched to SSS sequence. But generation of the report perturbed. It takes all the files to generate the report.

	Wednesday _28_A

Thurday_29_N
	TBA2
	3189 dipole HV test failed “dipole-QH- vs ground” at 3050 V

Happened again at the end of the test program


	MTM
	HV
	Repeated by OP team and by Richard: passed at the end.

A.Siemko and Richard decided to give the OK for the magnet.



	Wednesday_28_N
	TBD1
	Vtap VT4 connected to the lead F is opened so the potaim-card Vclamp F is triggering. Vtaps not properly soldered?
	ICS?
	
	Solved by Federico:bridge in the output of the card done by Federico between VT4 and VT3.

	Wednesday_28_N

Thursday_29_A
	TBD1
	Cryo OK cannot be obtained. 
	ACR
	
	Temp problem on auxiliary leads.

Potentiometer installed to keep the valve opened. But still not working.

 Lionel Herblin called by Federico. 

Manual mode needs more parameter than the automatic mode. Problem solved.

	Thursday _29_M
	TBA2
	SPA happened at 6500 A during the training 
	ACR-AB/CO
	
	Origin not known: Communication problem between PLC and CTH?

	Thursday _29_A
	TBC
	Water problem. 
	
	
	Other valves closed and then interlock disappeared. Origin not found. 

NB: Bench consumption: 80 l/min.

Total available: 440 l/min.: So 4 benches could work in parallel in theory. 

	Thursday _29_A
	TBE


	IAP : Offset on Potaim card D2_U-L at 5 V: 
	MTM
	
	Federico came and changed the card.

	Friday_30_A
	TBD1
	SSW measurements at 5000 A : During ramping down from 5000 A to 150 A, a water interlock appeared.

SPA abort at 5000 A during SSW.
	ACR-AB/CO
	
	Water closed for other benches.

Origin not found. Test repeated and then OK.

	Friday_30_N
	TBF1
	Water fault on PC during quench at 1.5 kA.
	
	
	Water flow kept only on two benches.

	Saturday_31_M

Satrurday_31_A

Saturday _31_N
	TBD1
	Gdl measurements with SSW on SSS23 (on total 11 times): 

-wire broke 4 times.

-SPA abort appeared on PC (2 times)

-After the measurements the wire was found at 

9mm below the centre in stage B : signal not good (4 times)

Wire centred but  TF value two low (signal too low).


	MTM

ACR-AB/CO
	
	Anticryostat problem? Wire touching the anticryostat.

Return wire not connected  to stage B

Diagnosis: R between wire and anticryostat : ~0.5 MOhm.

	Saturday_31_M
	TBC
	IAP at cold : Problem with the intercon box. Switch not possible.
	MTM
	
	Federico investigated the problem. Signal coming to the PLCSE not correct because the positioning of the auxiliary switch was not correct.

	Saturday_31_N
	TBE2
	Magnet 2035 : Min NRJ for warm-up: AQA error D1_U peak voltage at 150 V.
	MTM
	AQA
	Inform TC.

	Saturday_31_N

Sunday_01_A
	TBD2
	IAP at warm on SSS36: No data for Vtaps.
	MTM
	IAP
	Not the same channel name as compared to SSS23. Hf and lf files did not exist for this magnet.

Federico created the config files.

	Sunday_01_A
	TBC1
	Diode test : AQA alarm : Miits above the upper limits (23 at 3 kA)
	MTM
	AQA
	

	Sunday_01_A
	TBD1
	Slow PA during ramp for fast de-excitation test.
	ACR-AB/CO
	
	Origin unknown








